Author's response to reviews

Title: Personalized versus non personalized computerized decision support system to increase therapeutic quality control of oral anticoagulant therapy: an alternating time series analysis.

Authors:

Isabelle Colombet (isabelle.colombet@hop.egp.ap-hop-paris.fr)
Alessandra Bura-Riviere (alessandra.bura-riviere@hop.egp.ap-hop-paris.fr)
Remy Chatila (remy.chatila@hop.egp.ap-hop-paris.fr)
Gilles Chatellier (gilles.chatellier@hop.egp.ap-hop-paris.fr)
Pierre Durieux (pierre.durieux@egp.ap-hop-paris.fr)

Version: 2  Date: 21 September 2004

Author's response to reviews:

Changes were made based on formatting changes requested:
- the "et al." statements in references were replaced by the end of the authors list
- Figure 1 was cropped and its title and legend removed from the figure to be added in the main manuscript.